Author's response to reviews

Title: No risk for negative effects on cholesterol levels with a commercially available glucosamine product in patients treated with simvastatin or atorvastatin.

Authors:

Robert Eggertsen (robert.eggertsen@vgregion.se)
Åke Andreasson (ake.andreasson@vgregion.se)
Lennart Andrén (lennart.andren@vgregion.se)

Version: 2 Date: 6 October 2011

Author's response to reviews:

BioMed Central Clinical Pharmacology

I am hereby sending our manuscript "No risk for negative effects on cholesterol levels with a commercially available glucosamine product in patients treated with simvastatin or atorvastatin" partly revised for your evaluation and possible publication in your paper.

A completion of our registration number of the Medical Products Agency, Sweden from the their European registration is done, also in the abstract. Besides, the abstract is completed with background details and in methods (statistics)

A title page is added in the manuscript.

References are written in line with the reference style in BMC Clin Pharmacol.

Yours sincerely

Robert Eggertsen MD, PhD